Author's response to reviews

Title: MicroRNA and mRNA expression profiling in rat acute respiratory distress syndrome

Authors:

- Chaoqun Huang (chaogh@okstate.edu)
- Xiao Xiao (xiao.xiao@okstate.edu)
- Narendranath Reddy Chintagari (narendranath.chintagari@yale.edu)
- Melanie Breshears (melanie.breshears@okstate.edu)
- Yang Wang (yang.wang@okstate.edu)
- Lin Liu (lin.liu@okstate.edu)

Version: 4 Date: 8 July 2014

Author's response to reviews: see over
Dear Dr. Sleiman;

Enclosed please find our second revision of the manuscript entitled “MicroRNA and mRNA expression profiling in rat acute respiratory distress syndrome”. We have made all of the corrections identified by the reviewer, Tamas Dolinay.

We thank the reviewer for his/her continued efforts in improving this manuscript.

Sincerely;

Lin Liu, Ph.D.
Professor